Title: Removal of a below knee plaster cast worn for twenty eight months: a case report

Authors:

Helen Ingoe (Helen_ingoe@hotmail.com)
Sarah Eastwood (sarah.eastwood@talk21.com)
David W Elson (davidelson@yahoo.com)
Claire F Young (claire.young@ncuh.nhs.uk)

Version: 4 Date: 17 November 2010

Author's response to reviews:

I have made the minor revision so that the Competing interests section now precedes the Consent section.

The case presentation describes the events that happened to this patient over a five year period. She was 40 at the time of her bunion surgery, but she was 45 when she came to have her cast removed which is how it came to be on for so long and this is clearly stated in the text. I have changed the first sentence to make this clearer. Thanks